Author’s response to reviews

Title: The role of AFB microscopy training in improving the performance of laboratory professionals: Analysis of pre and post training evaluation scores

Authors:

Pawlos Reji (preji@msh.org)
Getachew Aga (getachewdoti@yahoo.com)
Gemeda Abebe (gemeda.abebe@ju.edu.et)

Version: 2 Date: 20 December 2012

Author’s response to reviews: see over
To: BMC Health Service Research

Dear Editors,

We, Pawlos Reji, Getachew Aga, and Gemeda Abebe have submitted our Original article titled ‘The role of AFB microscopy training in improving the performance of laboratory professionals: Analysis of pre and post training evaluation scores’ to BMC Public Health for possible publication. We believe that BMC Public Health is the right journal to disseminate our findings to researchers and policy makers timely.

Regards,

Gemeda Abebe (MSc), Assistant Professor.

Corresponding author.